

### Second job holding in Germany – a persistent feature? Lentge, Philipp

Publication date: 2022

**Document Version** Publisher's PDF, also known as Version of record

Link to publication

*Citation for pulished version (APA):* Lentge, P. (2022). Second job holding in Germany – a persistent feature? (University of Lüneburg Working Paper Series in Economics; Vol. 416). Institut für Volkswirtschaftslehre der Universität Lüneburg.

### **General rights**

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
  You may not further distribute the material or use it for any profit-making activity or commercial gain
  You may freely distribute the URL identifying the publication in the public portal ?

Take down policy If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

# *NORKING PAPER*

# Second job holding in Germany – a persistent feature?

by Philipp Lentge

University of Lüneburg Working Paper Series in Economics

No. 416

November 2022

www.leuphana.de/institute/ivwl/working-papers.html ISSN 1860 - 5508

### Second job holding in Germany — a persistent feature?

Philipp Lentge<sup>a</sup>

Abstract: This paper investigates the persistence and determinants of second job holding in Germany, especially marginal second jobs, following a legislative change allowing extensive dispensation of marginal second jobs from taxes and social security contributions. I document an upward trend in second job holding driven in particular by women. Moreover I find strong evidence for the persistence of second job holding using a dynamic panel model. Further, I identify significant gender differences in the decision to moonlight, where especially women in the lower part of the earnings distribution and women with few years of tenure are most likely to take up a second job.

JEL-Classification: J22; J28

Keywords: second job, marginal second job, multiple job holding

<sup>&</sup>lt;sup>a</sup> Philipp Lentge, Leuphana University of Lüneburg, Universitätsallee 1, 21335 Lüneburg, Germany, philipp.lentge@leuphana.de

## 1 Introduction

Second job holding, also referred to as moonlighting, is a notable labour market feature and is the topic of numerous studies documenting substantial numbers of the work force engaging in more than one job in most industrialized countries (recent studies include Auray *et al.*, 2021, and Choe *et al.*, 2017).

Standing out among other Western countries is Germany, where examining German data over the recent almost two decades suggests an unseen upward swing in second job holding, more than doubling in numbers and being the exception in comparison to other European countries (Klinger and Weber, 2020). This unique trend started after a change in the German legislation allowed for extensive dispensation of marginal employment, including marginal second jobs, from taxes and social security contributions starting in 2003. Figure 1 illustrates this strong increase and shows that in 2019 more than 2.5 million employees in Germany held at least one additional job, translating into about 9% of the employed with at least one job subject to social security. As noted by Klinger and Weber (2020) evidence for the German labour market is scarce and somewhat outdated (Heineck, 2009, and Schmidt and Voss, 2014)<sup>1</sup>, regarding the increase of second job holding over the last 15 years.

Turning to the existing literature, in general the reasons for holding multiple jobs can be characterized by two main motives. The so-called hours constraint (or earnings constraint) motive applies to workers being unable to supply their desired number of hours in a first job, which leads them to taking up an additional job to optimize their individual labor supply (e.g. Shishko and Rostker, 1976; Conway and Kimmel, 1998). Early research found that specifically workers' earnings on the first job determined the propensity to moonlight and concluded that people foremost take up additional jobs to increase living standards and meet financial needs (e.g. Hamel, 1967; Guthrie, 1969), as for example labor market institutions constrained the hours in their primary occupation (Perlman, 1966). Furthermore, Boheim and Taylor (2004) showed that higher income

The study by Heineck (2009) uses SOEP data up to the year 2005 and Schmidt and Voss (2014) use SOEP data up to 2011.



Figure 1: Share and number of employed having a second job (SOEP 2004-2019; weighted using sample weights)

levels in the first job decreased the propensity to take up a second job and also found that second jobs are not only temporary labor supply adjustments but rather persist over time.

Alternatively, the heterogeneous jobs (or portfolio) motive accounts for workers who do not face an hours constraint on their first job, but choose to work a second job for reasons beyond earnings. Workers may increase their utility from different aspects of the second job then just income alone, since there exist complementarities with the first job, e.g. the professor who also works as a consultant or a mother who holds two jobs to meet time-allocation needs (Heineck, 2009). Moreover workers may aim at diversifying their tasks to increase satisfaction by engaging in more pleasant or prestigious second jobs (Conway and Kimmel, 1998, and Renna, 2006) or invest in human capital (Averett, 2001). Empirical work supporting the portfolio motive includes Panos *et al.* (2014) showing that the acquired skills from the second job increase the probability of finding a new job or Pouliakas (2017) finding high-skilled moonlighters upgrading their skills. Auray *et al.* (2021) find that second job holding is rising with individuals' education and interpret their results as evidence for a comparative advantage of educated workers in second job holding. Allen (1998) (for the US) and Boheim and Taylor (2004) (for the UK) obtain that constrained workers are actually less likely to hold an additional job than unconstrained workers, thus suggesting that heterogeneous job motives are more important than the hours constraint. Other studies exploring the determinants of multiple job holding include Kimmel and Conway (2001), Wu *et al.* (2009) and Dickey *et al.* (2011).

Investigating economic conditions and moonlighting, Amuedo-Dorantes and Kimmel (2009) find for the US that multiple job holding of men is largely acyclic, while female multiple job holding switched to being procyclical in the early 2000s. Similarly, Hirsch *et al.* (2017) document that multiple job holding is largely acyclic and negligible in mitigating earnings volatility over the business cycle. Moreover it is rather unresponsive to unemployment or employment growth changes within markets over time.

Against this backdrop, this paper contributes to the existing literature as follows. Firstly, using recent panel data for Germany I document developments in second job holding since the legislative change in 2003 with a focus on marginal second jobs and how these vary especially between men and women. Secondly, I show empirically that second job holding is a persistent feature of workers' biographies in Germany. Thirdly, I investigate the individual determinants of taking up a second job—either a marginal or non-marginal one—and whether there are marked differences among specific groups of workers.

The remainder of this paper is organised as follows. Section 2 describes my data and provides descriptive findings on the prevalence and extent of second job holding in Germany. Section 3 explains the econometric approach based on a dynamic multinomial logit random effects model. Section 4 presents and discusses my estimation results, and Section 5 concludes.

### 2 Data and descriptive findings

Within the German labour market there is the distinction into two different types of employment: regular jobs subject to income tax and social security contributions, and marginal jobs (called mini-jobs), which are exempt from income taxes and the employee share of social security contributions (health, long-term care, unemployment, and pension) leading to 'gross is net' pay for those jobs. From 1999 to 2003 the social security exemption for marginal employment additional to the main job had been abolished and when holding a second job, these earnings were added to the main job earnings and were thus subject to income tax and social security contributions. The law governing marginal employment was changed with the Hartz II reforms, effective April 1, 2003, and introduced a marginal earnings threshold of 400 Euros and removed the limit of 15 hours per week for marginal employment and allowed workers in a regular main job to hold an additional marginal second job free of tax and social security contributions. As a result of legislative changes in the area of marginal employment, the monthly earnings limit for marginal employment was raised from 400 to 450 Euros as of January 1, 2013. At the same time, these minijobs became subject to compulsory insurance in the statutory pension insurance, but the option to be exempt was created. Thus, second job holding in Germany can be divided into holding a marginal (400/450 Euros per months up to 2013/from 2013 onwards) or nonmarginal second job (exceeding 400/450 Euros per month). Therefore, I use the earnings information on second jobs in my data to define these second jobs according to these thresholds, meaning into marginal and non-marginal second jobs.

My data come from the German Socio-Economic Panel (SOEP henceforth) for the years 2003<sup>2</sup> to 2019 (Goebel *et al.*, 2018). This survey is nationally representative and provides detailed information on individuals and households on an annual basis. The same individuals are re-interviewed each year, and if they leave their original households to form new households all adult members of these new households are also interviewed. I include only employees that have a primary job, which is subject to social security contributions, meaning this income must exceed 400 (450) Euros per month for the years up to 2013 (after 2012). Furthermore the sample is restricted to workers aged 18 to 65 who are full- or part-time employed on the first job. I exclude self-employed<sup>3</sup>, apprentices, interns, and partial retirees.

<sup>&</sup>lt;sup>2</sup> I only include observations for the year 2003 with the interview time being later than April 1st, since from that time on the legislative change was in effect.

<sup>&</sup>lt;sup>3</sup> I exclude self-employed in the main job as they have more flexibility of choosing their working hours and other aspects of their job.

	Main j	ob only	Margin	nal 2nd	Non-mar	ginal 2nd
Main job characteristics	Men	Women	Men	Women	Men	Women
Main job monthly gross income in $\in$	3,332.59	2,209.68	3,258.94	1,889.72	4,309.28	2,445.93
Decile of first job gross income $(1/0)$ :	,	,	,	,	,	,
Up to 1st decile	0.025	0.166	0.030	0.251	0.023	0.169
1st to 2nd decile	0.044	0.152	0.043	0.175	0.040	0.119
2nd to 3rd decile	0.077	0.141	0.065	0.156	0.048	0.116
3rd to 4th decile	0.092	0.108	0.092	0.111	0.041	0.116
4th to 5th decile	0.109	0.099	0.115	0.085	0.069	0.083
5th to 6th decile	0.115	0.090	0.138	0.065	0.083	0.077
6th to 7th decile	0.125	0.084	0.128	0.062	0.118	0.075
7th to 8th decile	0.131	0.073	0.136	0.046	0.112	0.089
8th to 9th decile	0.141	0.056	0.138	0.034	0.165	0.092
Above 9th decile	0.141	0.031	0.117	0.015	0.301	0.064
Tenure $(1/0)$						
Less than 2 years	0.155	0.177	0.156	0.249	0.134	0.196
2-5 years	0.167	0.185	0.175	0.217	0.158	0.208
5-10 years	0.178	0.185	0.179	0.204	0.183	0.210
More than 10 years	0.500	0.453	0.489	0.330	0.525	0.387
Full-time $(1/0)$	0.950	0.527	0.891	0.385	0.842	0.413
Temporary contract $(1/0)$	0.193	0.021	0.185	0.200	0.205	0.233
Public service sector $(1/0)$	0.225	0.357	0.331	0.200 0.357	0.400	$0.200 \\ 0.452$
Worker characteristics	0.220	0.001	0.001	0.001	0.100	0.102
Family status $(1/0)$						
Single and no children	0 184	0 188	0.198	0.226	0 134	0 188
Partner and no children	0.366	0.409	0.331	0.337	0.356	0.100
Partner and children	0.300	0.318	0.351	0.323	0.000	0.319
Single and children	0.017	0.085	0.014	0.114	0.016	0.091
Preferred working hours first job $(1/0)$	0.011	0.000	0.011	0.111	0.010	0.001
More hours	0.261	0.257	0.323	0.396	0.361	0.347
Less hours	0.292	0.319	0.282	0.254	0 294	0 294
Same hours	0.447	0.010 0.424	0.395	0.350	0.345	0.358
Employment status partner $(1/0)$	0.111	0.121	0.000	0.000	0.010	0.000
No working partner	0.517	0.526	0.511	0.608	0.526	0.604
Full time	0.194	0.446	0.175	0.368	0.147	0.376
Part time	0.239	0.020	0.232	0.017	0.282	0.016
Other	0.051	0.008	0.082	0.008	0.045	0.005
Age $(1/0)$	0.001	0.000	0.002	0.000	0.010	0.000
18-25 years old	0.046	0.049	0.044	0.048	0.018	0.013
26-35 years old	0.187	0.178	0.200	0.180	0.119	0 155
36-45 years old	0.301	0.303	0.316	0.302	0.330	0.305
46-55 years old	0.308	0.324	0.316	0.354	0.334	0.377
Older than 55 years	0.157	0.146	0.125	0.116	0.001	0.011
Skill level $(1/0)$	0.101	0.110	0.120	0.110	0.100	0.100
Low skill	0.074	0.072	0.074	0.073	0.029	0.038
Medium skill	0.660	0.649	0.628	0.682	0.020	0.000 0.457
High skill	0.266	0.279	0.298	0.002	0.490	0.505
Lives in East Germany $(1/0)$	0.200	0.210 0.247	0.136	0.154	0.100	0.000
Migration Background (1/0)	0.210	0.211	0.100	0.101	0.112	0.101
No migration background	0.815	0.838	0.806	0.824	0.891	0.862
Direct migration background	0 139	0.117	0.146	0.130	0.076	0.091
Indirect migration background	0.046	0.045	0.048	0.046	0.033	0.047
		0.010		0.010		110.0
Number of Individuals	13,984	13,809	1,535	1,916	595	367
Number of Observations	60,654	57,994	3,209	4,036	1,258	639

Table 1: Variable means by second job status

Number of Observations 60,654 57,994

Notes: SOEP 2003-2019; unbalanced panel, unweighted.

Table 1 tabulates the variable means for the estimating sample by second job status. The final sample comprises 127,790 observations (49% Women). The SOEP data include information on workers' personal and job characteristics. Moreover, the SOEP includes information on the household and family context of interviewed persons allowing me to include this information in later investigations. Workers characteristics comprise, inter alia, their sex, age, information on educational attainment, family status, employment status of partner, migration background, and whether the person lives in East or West Germany. Characteristics of the main job include workers' earnings, job tenure, temporary (as opposed to permanent) contract, working hours, whether the job is in the public service sector, occupation, industry sector, and firm size. For my further analysis I will make use of monthly income deciles (derived from monthly earnings across all observations by year) to allow for non-monotone income effects in latter models, as well as for descriptive evidence for different income groups. For workers' preferences regarding working hours, I can generate an appropriate indicator. SOEP-interviewees are asked: 'If you could choose your own working hours, taking into account that your income would change according to the number of hours: How many hours would you want to work?'<sup>4</sup>. Comparing this number of desired working hours with actual hours worked, gives me an indicator for the working hours preference of a worker.<sup>5</sup>

As is visible from Table 1, there are substantial differences between men's and women's characteristics and their second job status. Firstly, there are marked differences in monthly gross income by job status and gender. Men's income in the main, respectively only job, is in all cases higher than their female counterpart's. Comparing the incomes of workers with regard to what type of second job they hold, it shows that workers with a marginal second job tend to have lower incomes in their first job than those having a non-marginal second job. Note further that female marginal second job holders are disproportionally found in the lower main job income deciles compared to the other two outcomes, with almost 60% being in the lowest four deciles. Among male non-marginal second job holders, top earners

 $<sup>\</sup>overline{4}$  This question is asked directly after questions relating to the first (main) job.

<sup>&</sup>lt;sup>5</sup> Information on desired working hours is not available for 2019. Therefore latter estimations include data from 2003 to 2018.



Figure 2: Male share and number of employed having a second job (SOEP 2004-2019; weighted using sample weights)



**Figure 3:** Female share and number of employed having a second job (SOEP 2004-2019; weighted using sample weights)

make up the largest group with almost 50% being in the top two deciles. Moreover, female as well as male moonlighters work less often full time than employees with only one job.

Turning to workers' personal characteristics, we see from Table 1 that the proportion of single mothers is higher for second job holders, whereas among male marginal second job holders the proportion of men with partner and children is highest. About one third of second job holders prefers more hours in their main job, which could be an indication of the hours constraint. The distribution of skill levels is quite similar for workers with only one job and marginal second job holders, whereas among non-marginal second job holders the share of low skilled is smallest. Apart from that, we can see that the share of second job holders is smaller in Eastern Germany.

In the following, using these data and the provided sample weights, I will descriptively investigate trends and heterogeneities in second job holding for Germany for the years  $2004^6$  to 2019 for various subgroups. Note that I will mainly focus on marginal second jobs in this descriptive analysis, since for those jobs legislative changes were introduced.



*Figure 4:* Development of second job types for men and women (SOEP 2004-2019; weighted using sample weights)

<sup>&</sup>lt;sup>6</sup> I start with the year 2004 as I am comparing absolute yearly numbers and for the year 2003 that number would be very small, due to the legislative change in April 2003 and only a small number of SOEP interviews for the year 2003 after April of this year.

Having seen the strong overall increase in second jobs in Germany, Figures 2 and 3 show these developments separately for men and women. Men exhibited in 2004 initially a higher number of moonlighters than women in absolute and relative terms, but have been subsequently overtaken by women. From 2004 to 2019, the share of women holding more than one job rose from around 4 to more than 10 percent, whereby the absolute number almost tripled. The increase in the share of moonlighting men was moderate from about 6 to about 8 percent, since the legislation on mini-jobs changed. The number of men holding a second job increased during this period by about half a million. In addition to that, there is a visible dip in male second job holding for the year 2010, the time of the financial crisis and sharp drop of economic growth in Germany.

 Table 2:
 Number of second job holders by second job type

	Men		Wo	omen	
Year	2004	2019	2004	2019	
Marginal second job	635,000	950,000	445,000	1,215,000	
Non-marginal second job	210,000	240,000	$105,\!000$	$165,\!000$	
Notes: SOEP 2004-2019; weighted using sample weights.					

However, as pointed out before, there are two different types of second jobs in Germany; marginal and non-marginal ones. Figure 4 illustrates the development of these different second job types by gender. It stands out, that in particular marginal second jobs of women have increased. This increase is almost threefold and by far the strongest one during the considered time period. Whereas women's non-marginal second job numbers have increased by about 50 percent, we can see that men's non-marginal second job holding remained about the same, with some ups and downs during these years. Table 2 gives an indication of the magnitude of moonlighting workers, and further shows that marginal second jobs substantially outnumber non-marginal ones. Taking into account the increased female participation rate and also the strong German economic performance for the second part of my observation period, these descriptive findings nevertheless appear very plausible in connecting these increases to the legislative changes. It also shows that non-marginal second employment is rather unaffected.

Next, I will take a closer look at those marginal second job holders and who they



**Figure 5:** Development of marginal second job holding by monthly income quartiles for women (SOEP 2004-2019; weighted using sample weights)



**Figure 6:** Development of marginal second job holding by monthly income quartiles for men (SOEP 2004-2019; weighted using sample weights)

are. Figure 5 shows the development of marginal second job holding of women divided into monthly first job income quartiles of the whole distribution (including both men and women). The number of marginal second jobs is trending upwards for all income quartiles, but is largest for the lowest income quartile by a large margin. In 2019 there are three and a half times as many female marginal second job holders in the lowest income quartile than were in 2004. For the second quartile the number increased by more than 150%, while it also doubled for female top earners. On the contrary, the numbers of male marginal second job holders (Figure 6) are quite similar for all income quartiles but the second quartile, where second job holding even seems to decline over the years.



**Figure 7:** Share of second job holders by type and income decile (SOEP 2003-2019; weighted using sample weights)

This impression is also borne out in Figure 7 that shows the shares of marginal and non-marginal second holders for income deciles by gender. The lower income deciles of women have the highest share of marginal second job holders with every tenth women in this group moonlighting, while this share is decreasing when moving up the income distribution. Within the lower half of the income distribution, the share of women in marginal second jobs is always higher than for men. About 6% of male workers in the first income decile also engage in a marginal second job, with another peak for the sixth decile. Overall, I can see that the incidence to hold a marginal second job (mostly) decreases along the income distribution no matter the gender of the worker. As seen before, nonmarginal second job holding plays a minor role with shares between 1 or 2 percent, with the exception being the top decile for men, where it is almost on par with marginal second job holding.

In addition to differences in second job holding for specific income groups, I look at the development of marginal second jobs for specific skill levels of workers. I distinguish high-skilled workers with an academic education, medium-skilled workers with a vocational training, and low-skilled workers with neither. Figure 8 displays that marginal second jobs of women of all skill levels more than doubled over the considered time span; showing a very similar upward trend. By contrast, we can see in Figure 9 that especially low skilled men increased their engagement in marginal second jobs. For the medium skilled (the largest skill group) there is almost no change noticeable across the years, while for the high skilled it rose by about 50 percent.

	Income decile										
Family status of women	1	2	3	4	5	6	7	8	9	10	Overall
Single	10.32	9.43	8.7	9.91	7.31	6.36	8.16	4.29	4.33	7.55	7.89
Partner	9.96	6.47	5.6	7.04	5.04	4.68	4.35	6.49	3.78	1.21	6.13
Partner and children	10.02	6.79	6.05	5.29	5.97	5.8	5.24	4.33	3.54	3.3	6.85
Single with children	10.77	8.97	10.52	11.69	6.5	5.74	6.23	7.16	1.9	2.7	8.93
Family status of men											
Single	7.01	6.15	4.69	5.25	5.61	7.45	4.91	4.14	4.6	2.53	5.24
Partner	6.97	4.74	3.06	3.8	4.17	5.49	4.85	4.1	4.02	3.31	4.26
Partner and children	4.61	6.08	4.12	5.68	5.85	5.11	5.26	6.52	4.65	4.11	5.18
Single with children	0.88	1.94	2.96	2.23	6.23	7.62	2.09	0.24	3.49	7.99	3.12

Table 3: Share of marginal second job holders by family status for women and men

Notes: SOEP 2003-2019; weighted using sample weights.

In order to take further advantage of the strength of the SOEP data, I will look at the composition of marginal second job holders with regard to their family status (Table 3). As already seen, the share of women with a marginal second job is higher than those of men no matter the family status. Single mothers and single women show the highest shares of marginal second job holders with 9 respectively 8% among them. This fact is



**Figure 8:** Development of marginal second job holding by skill level for women (SOEP 2004-2019; weighted using sample weights)



**Figure 9:** Development of marginal second job holding by skill level for men (SOEP 2004-2019; weighted using sample weights)

even exacerbated when considering the lower income deciles. Further, we see that workers having children with a partner present a higher share of marginal second job holders than those without children. Taking a closer look at the different income deciles shows that the share of marginal moonlighters follows a downward trend while income rises across all groups.<sup>7</sup> All of these findings let me conclude so far, that when workers' conditions are below average or less than ideal, regarding e.g. income or household composition, marginal second jobs are particularly prevalent.

	Only main job	Marginal second job	Non-marginal second job	Overall
Men				
Hours main job	42.9	41.8	43.2	42.9
Hours in second job		5.4	8	6.1
Women				
Hours main job	35	32	33	35
Hours in second job		5.5	8.5	5.9
Notes SOED 2002 20	10. unmointe	4		

Table 4: Hours worked per week by job combination

Notes: SOEP 2003-2019; unweighted.

In addition, I want to make use of the data to show the allocation of working hours between the main and the second job. Firstly, Table 4 tabulates the mean hours worked per week and job combination. Those who have a marginal second job work on average fewer hours in their main job than those without a second mini-job, which provides some suggestive evidence for the hours constraint motive of moonlighting for this group. Here, women with a marginal second job work 3 hours less on average in their first job than women with only one job. Regardless of gender, workers with a marginal second job allocate on average 5.5 working hours per week to this second job. Men work more hours in their main jobs than women and seem to reduce these hours by not as much as women when holding an additional mini-job. Non-marginal second job holders work on average more hours in the main job than those with a marginal second job, whereas male nonmarginal second job holders seem to be workers with higher working hours across the

 $<sup>\</sup>overline{7}$  Single fathers are the exception, but are only very few in numbers in the data

board compared to the other types. This lends some evidence to the fact that particularly men with higher paying (more hours intensive) main jobs hold additional jobs with higher additional earnings opportunities.

As one of the main questions of this paper is the persistence of second job holding, I look at the transition matrix for second job holding. Looking at Table 5 the data seem to confirm this persistence. About 60% of workers that had a marginal second job in one year also had one in the next year. Also, about 54% of non-marginal second jobbers also had one in the next year. Apart from that, very few workers (about 3%) start any type of second job from year to year. Furthermore, it rarely occurs that workers switch from a marginal to a non-marginal second job in the next year. On the other hand, the opposite holds true for about 20% of non-marginal second jobbers. If we were to look at gender differences, all these numbers are somewhat larger for women than men.

		Period t	
	Only main job	Marginal second job	Non-marginal second job
Period t-1			
Only main job	97.25	2.27	0.48
Marginal second job	35.85	59.01	5.14
Non-marginal second job	27.22	19.06	53.72
Total	92.51	5.91	1.57

Table 5: Transitions between main job and second job combination types

*Notes*: SOEP 2003-2019; unweighted data; 162,577 observations from 33,747 individuals; figures indicate row percentages.

In summary so far, my descriptive findings show a strong increase in marginal second jobs in Germany. They further document significant differences in marginal second job holding across different socio-economic worker groups, where especially low income and low skill workers hold a marginal second job. What is more, there is descriptive evidence for a considerable persistence of second job holding. In a next step, I will run regressions to quantify the contribution of specific characteristics to the persistence and take-up of a second job.

### **3** Econometric Approach

In a first step, I investigate in a binary response model I investigate which characteristics are correlated with starting a marginal second job. I focus on marginal second jobs only, because no meaningful regression analysis is feasible for non-marginal second jobs due to the very low number of people starting a non-marginal second job in my sample. Moreover, the legislative change only affected marginal second jobs. Since taking up such a marginal second job is rare, I choose a complementary log-log model, which is asymmetrical and thus suitable for modelling such rare events.

In a next step, my econometric approach to study persistence rests on a dynamic random-effects logit model. I translate the decision to hold a second job into a multinomial logit setting to distinguish between types of second jobs and explore the factors driving a worker's decision to have either a main job, the main and a marginal second job, or a main and a non-marginal second job. Using a dynamic specification accounts for the state dependence of workers, modeled as a t - 1 lag in my outcome variable.

In such a model, identifying true state dependence rests on the assumption of no correlation between unobserved heterogeneity and the outcome variable. Another issue is the initial condition problem stemming from the correlation between the relevant unobserved factors and the initial observation  $y_{i0}$ . The initial period  $y_{i0}$  in my data regarding second jobs might not (and likely does not) correspond to the beginning of the stochastic process leading to the second job engagement. Therefore, the approach of the model I am using is based on the simple solution to the initial condition problem by Wooldridge (2005) and the extension by Rabe-Hesketh and Skrondal (2013) and means that I include the initial value of the response variable and of other time-varying explanatory variables.

Following this, my model has the following specification:

$$y_{it} = \gamma \mathbf{Z}_{it} + \rho y_{it-1} + c_i + u_{it} \tag{1}$$

The outcome variable  $y_{it}$  expresses the chances of holding a particular job type combination for worker *i* at time t.<sup>8</sup>  $\mathbf{Z}_{it}$  is a set of time-varying explanatory variables considered strictly exogenous, conditional on the unit-specific unobserved effect  $c_i$ .  $\rho$ captures the coefficient for the  $y_{it-1}$  and thus true state dependence, while  $u_{it}$  is an error term.

As Rabe-Hesketh and Skrondal (2013) note, the unit-specific unobserved effect  $c_i$  can be written as:

$$c_i = \alpha_0 + \alpha_1 y_{i0} + \overline{\mathbf{Z}}_i \alpha_2 + \mathbf{Z}_{i0} \alpha_3 + a_i \tag{2}$$

where  $\mathbf{Z}_{i0}$  and  $y_{i0}$  represent the initial values of the time-varying explanatory variables and of the response variable, respectively.<sup>9</sup>  $\overline{\mathbf{Z}}_i$  comprises the within-unit averages of the explanatory variables where the averages are based on all periods. Lastly,  $a_i$  is a unitspecific time-constant error term, normally distributed with mean 0 and variance  $\sigma_a^2$ . If the assumption holds, that unobserved heterogeneity is captured by  $c_i$ , then the lagged value of the response variable can be interpreted as true state dependence, meaning as the effect of holding a second job in one period on the second job status in a subsequent time period.

As covariates I add the following groups of variables: (i) main job characteristics including groups of dummies for income deciles, tenure, full-time hours, a temporary contract, public-service sector, as well as dummies for occupation, sector, and firm size; (ii) worker characteristics capturing family status, hours preferences, educational attainment (distinguishing high-skilled workers with an academic education, medium-skilled workers with a vocational training, and low-skilled workers with neither), employment status of the partner, age, migration background, and residence in East Germany. I employ dummies for the wage deciles as an explanatory variable to allow for non-monotone income effects suggested by the descriptive analysis in Section 2. I further control for working time arrangements (full- vs part-time) to disentangle both effects.

<sup>&</sup>lt;sup>8</sup> Note that t = 0 indicates the initial period.

<sup>&</sup>lt;sup>9</sup> They also show that this can be safely implemented in case of an unbalanced panel.

### 4 Estimation Results

First, I show the results for characteristics which correlate with workers to take up a marginal second job (Table 6).<sup>10</sup> Therefore I use a complementary log-log model to investigate the workers that take up a marginal second job in period t after working only their main job in t - 1 and note in passing that my findings remain robust when using a standard logit model.

At first, I discuss female workers. Having controlled for other influences like family status or working full-time, the probability of starting a marginal second job is by far the highest in the lowest income and lowest tenure group. Compared to the top 10% of high-earners, the lowest income decile workers are 2 percentage points more likely to hold a marginal second job. This translates into an about one third increase in the average probability to have a marginal second job and thus by no means a small effect as the overall average marginal second job rate form women is around 6% in the sample. The same effect size can be seen for women with less than 2 years of tenure compared to women with tenure of 10 years and more. Furthermore income growth has an impact, since moving up in the income distribution shrinks the probability of a marginal second job slowly but steadily, as staying longer with an employer does as well. This is consistent with the results of Klinger and Weber (2020) that document a similar pattern and conclude no evidence for the heterogeneous/ portfolio motive of moonlighting, since financial needs seem to dominate.

Other things being equal, full-timers and women with a partner are less likely to take up a marginal second job. The fact, that a woman wants to work more hours instead of the same number of hours in the main job increases the probability to take up a marginal second job by 1.4 percentage points, and further points at a possible hours, respectively earnings, constraint motive for the second job. Age plays a minor role for women to look for a marginal second job, as I can only see that women over the age of 55 are marginally less likely to do so. Living in East Germany reduces the average probability to engage in a marginal second job by 25% or 1.3 percentage points, which is highly likely due to a

<sup>&</sup>lt;sup>10</sup> See Appendix A.1 showing coefficients and standard errors.

Table 6:	Complementary log-log model for holding a marginal second job after only main
	job in period t-1; marginal effects

	Women	Men
Main job characteristics		
Main job monthly gross income		
up to 1st decile	0.019***	0.008
1st to 2nd decile	0.014***	0.013**
2nd to 3rd decile	0.014***	0.008**
3rd to 4th decile	0.012**	0.009**
4th to 5th decile	0.011**	0.012***
5th to 6th decile	0.001	0.012
6th to 7th decile	0.003	0.005
7th to 8th decile	0.001	0.000
8th to 9th decile	0.000	0.007
above 0th decile (reference)	0.004	0.002
Tenune		
leng there 2 means	0.000***	0.001
less than 2 years	0.022	-0.001
2-5 years	0.011***	-0.001
5-10 years	0.008***	-0.003
more than 10 years (reference)		
Full-time $(1/0)$	-0.006***	-0.007**
Fixed-term contract $(1/0)$	0.003	0.002
Public service sector $(1/0)$	0.002	0.006**
Personal characteristics		
Family status		
Single and no children (reference)	_	_
Partner and no children	-0.007***	0.001
Partner and children	-0.004	0.002
Single and children	0.001	0.002
Preferred working hours first job		
More hours	0.014***	0.006***
Same hours (reference)		
Less hours	0.002	0.001
Employment status partner		
Full time (reference)		
Not working	0.008***	0.001
Part time	0.01	0.002
Other	0.022**	0.011***
Age	0.022	0.011
	0.004	0.014**
26-35 years old	0.009***	0.013***
36-45 years old	0.007*	0.008***
46-55 years old	0.009***	0.005***
older than 55 years (reference)		
Skill loval		
	0.006*	0.006**
LOW SKIII	-0.000	-0.000
$\mathbf{MECHUMII SKIII} (\text{reference})$	0.007***	0.002
$\begin{array}{c} \text{High Skill} \\ \text{Lives in Fast Commercy (1/0)} \end{array}$		-0.002
Lives in East Germany (1/0)	-0.013	-0.010
Nigration Background		
No migration background (reference)		
Direct migration background	0.004	-0.001
Indirect migration background	-0.005	-0.004
	44.476 obs./	48.086 obs /
	10.679 persons	11 045 persons

 $\left|\begin{array}{c}10,679 \text{ persons} \\11,045 \text{ persons} \\ \text{Notes: SOEP 2003-2018. Standard errors clustered at the person level (* p<0.05, ** p<0.01, *** p<0.001). Controls included for industry sectors, occupation, firm size and years. \end{array}\right|$ 

higher share of full-time working women than in Western Germany.

Table 6 shows that men in the middle of the income distribution are most likely to start a second job, e.g. men in the fifth income decile have a 1.2 percentage points or 35% higher probability to have a marginal second job than top male earners. Moreover, men working in the public service sector have a 15% higher average probability to engage in a marginal second job compared to those in the private sector. This might be explained by hours and job security in the public sector job. I also find that the desire to work more hours, as well as being in prime working age, increases the chances to engage in an additional mini-job. As was the case for women, living in East Germany reduces the probability to moonlight.

In summary, I conclude that in particular a less than ideal first main job environment (income, tenure, part-time) of women appears to be a good indicator of taking up an additional mini-job.<sup>11</sup> For men the same can be said for average income levels and working in the public-service sector. On the one hand, the hours constraint motive or probably better earnings constraint motive gets support by the fact that workers with a preference to work more hours show a higher probability to look for an additional marginal job, where gross pay is net pay. On the other hand, the portfolio motive is hardly confirmed, as the probability to hold a second job does not rise significantly towards the top of the wage distribution.

Next, in Table 7 I present the marginal effects of the dynamic multinomial logit model with random effects for different second job status types of German workers.<sup>12</sup> The main variables of interest are the lagged second job states. Once controlled for the initial condition and net of the role of unobserved heterogeneity, these coefficients indicate the presence of significant genuine state dependence.<sup>13</sup>

Holding only the main job in t-1 serves as the reference category, such that the

<sup>&</sup>lt;sup>11</sup> This is in line with the findings of Schmidt and Voss (2014) for the period up to 2010.

<sup>&</sup>lt;sup>12</sup> The sum of the marginal effects for the alternatives: marginal second job and non-marginal second job, yield the respective marginal effect of the alternative: only the main job.

<sup>&</sup>lt;sup>13</sup> See Appendix A.2 showing coefficients and standard errors.

reported marginal effects must be interpreted relative to this group. I observe a genuine state dependence for second job holders, where the state dependence is even more pronounced for women: having a marginal second job in year t-1 increases the probability of holding a marginal second job in year t by about 17 percentage points or 200% for women and 11 percentage points for men (keeping in mind that the average probability of holding a marginal second job is about 6% for women and 4.4% for men across my sample). The effect of having a non-marginal second job in t-1 on having a marginal one the year after is almost equal in size and also highly significant. Having a marginal second job or, respectively, a non-marginal second job in t-1 also increases the probability to hold a non-marginal second job in t by 3, or 5 percentage points respectively. The same pattern can be found for men, but the effect of state dependence regarding marginal second jobs is only about two thirds of the effect size of women. These findings strongly support the argument that second job holding is no temporary but a persistent feature, in particular for German female workers with marginal second jobs.

Note that in this dynamic setting, genuine state dependence becomes the single largest (and most significant) contributor to the probability for holding some type of second job. Nevertheless, I want to comment on the impact of other variables as well, which are unfortunately not estimated very precisely due to the small number of observations for this outcome. Firstly regarding marginal second jobs, the probability to have a marginal second job is highest for the lower income deciles for women as well as men. For women, tenure of less than 2 years increases the probability of a marginal second job compared to longer tenured workers, while the desire for more working hours in the first job is associated with a higher probability to have a marginal second job, as expected. Moreover low skilled women show an 11 percentage points higher probability to hold a marginal second job than medium skilled women, while high skilled females are also more prone to hold a marginal second jobs than medium skilled ones. Meanwhile low skilled men are less likely to hold a second job than men with vocational training. Apart from that, living in East Germany reduces the probability for men to hold a marginal second job by almost 5 percentage points. Therefore, I can conclude that for my sample of German workers

	, T	Women		Men
	Marginal	Non-marginal	Marginal	Non-marginal
State dependence t-1				
Only main job (reference)				
Marginal second job	0.172***	$0.032^{***}$	0.112***	0.030***
Non-marginal second job	0.165***	0.047**	0.108***	0.041***
Main job characteristics				
Main job monthly gross income				
Up to 1st decile	0.016	0.004	$0.028^{*}$	0.013
1st to 2nd decile	0.010	0.002	0.014	$0.015^{*}$
2nd to 3rd decile	0.007	0.001	0.009	0.010
3rd to 4th decile	0.007	0.002	0.011	-0.001
4th to 5th decile	0.004	0.000	$0.016^{**}$	0.006
5th to 6th decile	-0.001	-0.001	0.009	0.001
6th to 7th decile	-0.006	-0.001	0.004	0.002
7th to 8th decile	0.001	-0.001	0.005	0.003
8th to 9th decile	-0.005	0.003	0.006	0.000
Above 9th decile (reference)	—			—
Tenure				
Less than 2 years	$0.014^{**}$	0.001	-0.005	-0.001
2-5 years	0.004	-0.001	-0.002	-0.002
5-10 years	0.001	0.001	-0.005	0.002
More than 10 years (reference)				
Full-time $(1/0)$	-0.010**	-0.005**	-0.008	0.000
Temporary contract $(1/0)$	0.002	-0.001	-0.002	0.003
Public service sector $(1/0)$	0.001	0.001	0.007	0.004
Worker characteristics				
Family status				
Single and no children (reference)		—		—
Partner and no children	-0.010	-0.001	-0.002	0.000
Partner and children	-0.007	-0.003	0.000	-0.004
Single and children	-0.003	-0.001	-0.007	$0.010^{***}$
Preferred working hours first job	0.000**	0.001	0.000	0.000
More hours	0.008**	-0.001	0.002	0.002
Same hours (reference)				
Less hours	0.001	0.002	-0.001	0.000
Employment status partner				
No working partner (reference)				
Full time	-0.008	-0.001	-0.000	0.000
Part time	-0.000	0.000	0.001	-0.001
Ama	0.005	0.000	0.007	-0.005
Age 18.25 years old	0.014	0.006	0.012	0.007
26, 25 years old	0.014	-0.000	0.015	0.007
20-35 years old $36.45$ years old	0.008	-0.004	0.005	-0.005
46.55 years old	0.000	-0.001	0.000	0.003
Older than 55 years (reference)	0.005	0.000	0.003	0.002
Skill lovel				
Low skill	0 119	-0.005	-0.023	-0 012***
Medium skill (reference)		_0.000		
High chill	0.033	0.015	-0.007	0.016
Lives in East Germany $(1/0)$	0.003	-0.020**	-0.046***	-0.005
Migration Background	0.000	0.020	0.010	0.000
No migration background (reference)				
Direct migration background	0.004	0.000	0.006	-0.004*
Indirect migration background	0.001	-0.001	0.000	-0.007**
			1	

### Table 7: Dynamic Multinomial logit model with random effects for second job states; marginal effects

-

.....

second job holding is persistent and even more so marginal second job holding and that especially for women in lower income groups, with low skills, and less job security (low tenure) the odds to have a marginal second job are highest. Descriptively, we had already seen that these groups showed the strongest increase in numbers across my sample period.

Overall, these estimations let me conclude that the probability of taking up a marginal second job depends a lot on individual and job characteristics — but once a worker is in, she is in, meaning that state dependence is very large for second job holders in Germany.

### 5 Conclusions

This study analyzes true state dependence in second job holding of German workers as well as the development of second job holding in Germany using panel data from the SOEP after the legislative change in 2003. This change allowed extensive dispensation of marginal second jobs from taxes and social security contributions. To that end, I analyzed the trends in second job holding overall and divided into the different types of second jobs.

Descriptively, I saw that women are more prone to hold second jobs and that the strong increase in numbers was mainly driven by marginal second jobs also known as mini-jobs. I further observed that the pronounced increase in marginal second jobs was largest in relative terms for low income women. Also, the data showed that non-marginal second job holders are a minority and seem to be unaffected by the legislative change as expected.

Using this panel data and taking into account the initial conditions problem, I estimate a dynamic multinomial logit model with random effects. I find that there is true state dependence in second job holding, especially for marginal second employment. This state dependence is even more pronounced for women than for men. Moreover, I find in panel estimations that low income, low tenure, part-time, and hours constrained women have a higher probability to take up a marginal second job. In short, my results show that gender differences in second job holding exist and that women in particular responded to the legislative change, which allowed for tax and social security contribution exempt second jobs.

Klinger and Weber (2020) argue that individual factors explain the increase over time

only to a very small extent and the tax exemption of marginal second jobs sets wrong incentives. I tend to follow their interpretation regarding wrong incentives of marginal second jobs and would see some further disadvantages. First, mini-jobbers themselves are losing out on certain social security benefits for their work in marginal second jobs, e.g. unemployment insurance or pension. This can be critical, as we have seen that especially low income groups tend to take up these jobs. Moreover, marginal second jobs set the incentive to stay below a certain threshold in earnings, although the individual might be able to work more. Secondly, it is unclear whether the tax exemption is desirable from a social point of view since the state forgoes tax revenue, as well as social insurance contributions. Even if the benefit of additional earnings capacities for low income workers is desired, the question remains, why there is the necessity for tax free second employment of well to do earners.

I expect the upward trend in marginal second jobs to go on, since the earnings limit for marginal employment has been recently increased from 450 to 520 euros per month. This could be used in future research to test how workers react in allocating their hours and how this continues to shape second job holding in Germany.

### A Appendix

	Women	Men
Main job characteristics		
Main job monthly gross income		
up to 1st decile	0.82***	0.50
•	(0.31)	(0.31)
1st to 2nd decile	0.67**	0.70***
	(0.30)	(0.25)
2nd to 3rd decile	0.68**	$0.48^{**}$
	(0.30)	(0.22)
3rd to 4th decile	$0.58^{*}$	$0.53^{***}$
	(0.30)	(0.20)
4th to 5th decile	$0.56^{*}$	$0.69^{***}$
	(0.30)	(0.19)
5th to 6th decile	0.45	$0.56^{***}$
	(0.30)	(0.18)
6th to 7th decile	0.21	0.38**
	(0.30)	(0.18)
7th to 8th decile	0.32	0.42***
	(0.31)	(0.16)
8th to 9th decile	0.25	0.16
	(0.31)	(0.15)
above 9th decile (reference)		
less than 2 years	0 78***	-0.064
	(0.10)	(0.13)
2-5 years	0.46***	-0.041
	(0.093)	(0.11)
5-10 years	0.35***	-0.14
•	(0.092)	(0.10)
more than 10 years (reference)		
Full-time $(1/0)$	-0.25***	-0.34**
	(0.095)	(0.15)
Fixed-term contract $(1/0)$	0.11	0.083
	(0.10)	(0.15)
Public service sector $(1/0)$	0.079	0.29**
	(0.084)	(0.11)
Industry		
Agriculture (reference)		
Energy	-0.63	-0.098
Minim	(0.57)	(0.55)
Millig		-0.08
Manufacturing	$(\cdot)$	(1.05) 0.27
Manufacturing	(0.36)	(0.38)
Construction	-0.24	-0.032
	(0.41)	(0.40)
Trade	-0.55	0.30
11000	(0.35)	(0.39)
Transport	-0.47	0.27
	(0.39)	(0.40)
Bank, Insurance	-0.39	0.53
<i>,</i>	(0.39)	(0.43)
Services	-0.34	0.57
	(0.35)	(0.38)
Other	-0.10	0.20
	(0.40)	(0.53)

Table A.1: Complementary log-log model for holding a marginal second job after only<br/>main job in period t-1

	Women	Men
Occupation		
Untrained worker (reference)		
Semi-Trained worker	-0.43**	0.11
	(0.18)	(0.28)
Trained worker	-0.49**	-0.0077
	(0.24)	(0.28)
Foreman, group leader	-0.53	0.28
	(0.59)	(0.33)
Foreman	-0.16	-0.31
	(0.77)	(0.45)
Industry or factory foreman	-0.49	-0.21
	(1.03)	(0.46)
Salaried employee, unskilled without training	-0.20	0.34
	(0.17)	(0.32)
Salaried employee, unskilled with training	-0.46***	0.11
r yr	(0.17)	(0.30)
Salaried employee, skilled	-0.49***	0.058
Salarioa Shipiojoo, Shihod	(0.17)	(0.28)
Salaried employee, highly skilled or managerial duties	-0.15	0.12
survive employee, inging skined of managenal duties	(0.20)	(0.30)
Solariad amployee with extensive managerial duties	(0.20)	0.70**
Satafied employee with extensive managerial duties	-0.23	(0.24)
	(0.41)	(0.54)
Civil servant, lower level	$\begin{pmatrix} 0 \\ \end{pmatrix}$	0.011
	(.)	(0.61)
Civil servant, middle level	-0.49	-0.35
	(0.33)	(0.37)
Civil servant, upper level	-0.81***	0.089
	(0.28)	(0.34)
Civil servant, executive level	-0.72**	0.72**
D'	(0.37)	(0.36)
r Irinsize		
20 to 100 omployoos	0.20***	0.15
20 to 199 employees	(0.084)	(0.12)
200 and more employees	0.004)	(0.12)
200 and more employees	(0.082)	(0.12)
Worker characteristics	(0.002)	(0.12)
Family status		
Single and no children (reference)		
Durther and no children	-0 28***	0.043
	-0.20	(0.12)
Dortnon and children	0.10	
r artifer and children	-0.13	(0.12)
Single and children	0.022	
Single and children	0.033	0.082
Proformed working hours first job	(0.11)	(0.30)
More hours	0 51***	0.21***
More nours	(0.01)	(0.00)
Como hours (noference)	(0.073)	(0.000)
Same nours (reference)	0.087	
Less nours	(0,000)	0.049
Employment status neutron	(0.082)	(0.088)
Employment status partner		
run time (reierence)	0.20***	0.022
Not working	$0.32^{+++}$	0.033
	(0.084)	(0.12)
Part time	$0.37^{*}$	0.13
	(0.21)	(0.12)
Other	0.70***	0.51***
	(0.26)	(0.16)

# Table A.1 continued:Complementary log-log model for holding a marginal second job<br/>after only main job in period t-1

	Women	Men
Age		
18-25 years old	0.20	$0.78^{***}$
	(0.19)	(0.25)
26-35 years old	0.37***	$0.73^{***}$
	(0.14)	(0.15)
36-45 years old	0.28**	$0.50^{***}$
	(0.13)	(0.15)
46-55 years old	$0.36^{***}$	$0.32^{**}$
	(0.12)	(0.13)
older than 55 years (reference) Skill level	_	
Low skill	-0.23*	-0.36**
	(0.13)	(0.17)
Medium skill (reference)		
High skill	-0.27***	-0.089
Ŭ	(0.10)	(0.12)
Lives in East Germany $(1/0)$	-0.59***	-0.70***
	(0.094)	(0.12)
Migration Background		
No migration background (reference)		
Direct migration background	0.15	-0.062
6 6	(0.097)	(0.13)
Indirect migration background	-0.20	-0.26
	(0.17)	(0.18)
Year		()
2004 (reference)		
2005	0.37	0.81**
	(0.32)	(0.38)
2006	0.31	0.63
	(0.32)	(0.40)
2007	0.45	1.09***
	(0.32)	(0.39)
2008	$0.54^{*}$	0.83**
	(0.32)	(0.39)
2009	0.72**	0.82**
	(0.32)	(0.39)
2010	0.30	0.54
	(0.33)	(0.41)
2011	0.71**	1.15***
	(0.33)	(0.39)
2012	0.69**	0.85**
	(0.32)	(0.39)
2013	0.81***	0.81**
	(0.31)	(0.39)
2014	0.83***	0.88**
	(0.31)	(0.39)
2015	$0.59^{*}$	0.92**
	(0.31)	(0.39)
2016	0.54*	1.06***
2010	(0.31)	(0.39)
2017	1.05***	1.62***
2011	(0.31)	(0.38)
2018	0.88***	1.19***
_010	(0.32)	(0.39)
	(0.04)	(0.00)

# Table A.1 continued:Complementary log-log model for holding a marginal second job<br/>after only main job in period t-1

Notes: SOEP 2003-2018. Standard errors in parentheses (\* p < 0.05, \*\* p < 0.01, \*\*\* p < 0.001).

	Women		Men		
	Marginal	Non-marginal	Marginal	Non-marginal	
State dependence t-1					
Only main job (reference)				—	
Marginal second job	$2.706^{***}$	$3.017^{***}$	$2.506^{***}$	$2.258^{***}$	
	(0.123)	(0.235)	(0.135)	(0.238)	
Non-marginal second job	$2.672^{***}$	$3.496^{***}$	$2.477^{***}$	$2.640^{***}$	
	(0.251)	(0.445)	(0.209)	(0.299)	
Main job characteristics					
Main job monthly gross income					
up to 1st decile	0.495	0.774	1.114**	$1.326^{*}$	
-	(0.427)	(0.933)	(0.483)	(0.709)	
1st to 2nd decile	0.317	0.425	$0.693^{*}$	1.348**	
	(0.412)	(0.835)	(0.402)	(0.599)	
2nd to 3rd decile	0.233	0.173	0.461	$0.979^{*}$	
	(0.401)	(0.806)	(0.356)	(0.561)	
3rd to 4th decile	0.239	0.415	0.464	0.0109	
	(0.392)	(0.742)	(0.325)	(0.593)	
4th to 5th decile	0.119	0.0997	0.684**	0.681	
	(0.384)	(0.735)	(0.308)	(0.476)	
5th to 6th decile	-0.0385	-0.242	0.424	0.173	
	(0.371)	(0.671)	(0.296)	(0.446)	
6th to 7th decile	-0.238	-0.277	0.200	0.255	
	(0.357)	(0.692)	(0.275)	(0.396)	
7th to 8th decile	0.0419	-0.236	0.256	0.321	
	(0.353)	(0.611)	(0.260)	(0.336)	
8th to 9th decile	-0.150	0.428	0.252	-0.00547	
	(0.314)	(0.581)	(0.218)	(0.278)	
above 9th decile (reference)	—			_	
Tenure					
less than 2 years	$0.389^{**}$	0.168	-0.233	-0.204	
	(0.175)	(0.401)	(0.218)	(0.372)	
2-5 years	0.155	-0.155	-0.0754	-0.168	
	(0.151)	(0.338)	(0.189)	(0.359)	
5-10 years	0.0514	0.204	-0.220	0.0839	
	(0.126)	(0.316)	(0.151)	(0.283)	
more than 10 years (reference)					
Full-time $(1/0)$	-0.340**	-0.892***	-0.363	$-0.784^{**}$	
	(0.152)	(0.329)	(0.261)	(0.379)	
Fixed-term contract $(1/0)$	0.0577	-0.111	-0.0467	0.283	
	(0.143)	(0.343)	(0.183)	(0.293)	
Public service sector $(1/0)$	0.0403	0.240	0.326*	0.423	
	(0.173)	(0.399)	(0.196)	(0.331)	
Occupation					
Untrained worker (reference)					
Trained worker	-0.0863	0.525	-0.128	-0.612	
	(0.171)	(0.526)	(0.248)	(0.376)	
Foreman, highly trained, managerial duties	0.0726	0.397	0.176	-0.234	
	(0.231)	(0.621)	(0.297)	(0.460)	

 $\label{eq:table A.2: Dynamic Multinomial logit model with random effects for second job states$ 

	v	Vomen		Men	
	Marginal	Non-marginal	Marginal	Non-marginal	
Industry					
Agriculture, energy, mining	0.783	-20.27***	0.555	-0.362	
righteureure, energy, inning	(0.749)	(1.285)	(0.574)	(0.741)	
Manufacturing	0.260	0.985	1.042**	0.0554	
	(0.374)	(0.815)	(0.468)	(0.689)	
Construction	0.424	5.105***	0.584	-0.0617	
	(0.517)	(1.309)	(0.521)	(0.777)	
Services	0.162	0.622	0.758*	-0.536	
	(0.366)	(0.716)	(0.452)	(0.705)	
Firmsize	~ /			( )	
less than 20 employees (reference)					
20 to 199 employees	0.144	-0.554	0.212	0.0874	
1 V	(0.161)	(0.424)	(0.219)	(0.347)	
200 and more employees	0.0377	-0.599	0.0891	0.0359	
1 V	(0.169)	(0.476)	(0.260)	(0.360)	
Worker characteristics Family status				· · · ·	
Single and no children (reference)					
Partner and no children	-0.315	-0.235	-0.0709	-0.0454	
	(0.192)	(0.357)	(0.230)	(0.360)	
Partner and children	-0.233	-0.574	-0.0261	-0.447	
	(0.218)	(0.455)	(0.233)	(0.362)	
Single and children	-0.0965	-0.205	-0.398	-1.343**	
	(0.220)	(0.525)	(0.514)	(0.559)	
Preferred working hours first job	(00)	(0.0_0)	(0.01-)	(0.000)	
More hours	0.234**	-0.187	0.103	0.175	
	(0.0928)	(0.219)	(0.111)	(0.172)	
Same hours (reference)	(0.0010)	(0.220)		(*****)	
Less hours	0.0528	0.323	-0.0573	-0.0220	
	(0.103)	(0.220)	(0.106)	(0.175)	
Employment status partner	(01100)	(00)	(01200)	(01110)	
Not working (reference)					
Full time	-0.246*	-0.259	-0.278	-0.0513	
	(0.142)	(0.283)	(0.175)	(0.273)	
Part time	-0.183	-0.0581	0.0481	-0.0693	
	(0.281)	(0.630)	(0.151)	(0.210)	
Other	0.107	0.736	0.234	-0.246	
	(0.401)	(1.204)	(0.193)	(0.290)	
Age	()			()	
18-25 years old	0.366	-1.415	0.550	0.756	
	(0.432)	(1.396)	(0.505)	(1.226)	
26-35 years old	0.215	-0.671	0.190	-0.315	
· - 0 · · · · · · · ·	(0.300)	(0.708)	(0.335)	(0.549)	
36-45 vears old	-0.0219	-0.172	0.0587	0.501	
	(0.224)	(0.541)	(0.246)	(0.402)	
46-55 years old	0.0935	0.0897	0.150	0.287	
			1 0.100		
40 00 years old	(0.160)	(0.394)	(0.184)	(0.262)	

	Women		Men		
	Marginal	Non-marginal	Marginal	Non-marginal	
		-			
Skill level					
Low skill	2.183**	-1.211	-1.291	-2.836**	
	(0.992)	(1.842)	(1.115)	(1.327)	
Medium skill (reference)					
High skill	1.008	$1.965^{*}$	-0.159	$1.279^{*}$	
	(0.800)	(1.021)	(0.608)	(0.736)	
Lives in East Germany $(1/0)$	0.0166	-3.525***	-2.365***	-0.899	
	(0.694)	(0.965)	(0.854)	(0.667)	
Migration Background					
No migration background (reference)					
Direct migration background	0.145	0.0931	0.214	-0.435	
	(0.129)	(0.338)	(0.163)	(0.309)	
Indirect migration background	0.0434	-0.165	-0.0838	-0.919**	
	(0.190)	(0.500)	(0.246)	(0.466)	
Year					
2004 (reference)					
2005	0.330	-0.177	0.430	0.310	
2004	(0.267)	(0.506)	(0.269)	(0.367)	
2006	0.140	-0.364	0.152	0.297	
	(0.278)	(0.553)	(0.303)	(0.437)	
2007	0.193	-0.676	0.440	0.657	
2000	(0.280)	(0.556)	(0.297)	(0.416)	
2008	0.318	-0.192	0.255	0.548	
2000	(0.282)	(0.531)	(0.306)	(0.431)	
2009	$0.494^{*}$	-0.158	0.310	0.582	
2010	(0.283)	(0.559)	(0.300)	(0.447)	
2010	(0.209)	-0.449	0.00409	0.440	
0011	(0.283)	(0.539)	(0.307)	(0.439)	
2011	(0.337)	-0.893	$0.524^{*}$	$0.881^{**}$	
2012	(0.290)	(0.579)	(0.312)	(0.424)	
2012	(0.350)	-1.140	(0.349)	(0.452)	
2012	(0.207)	(0.309)	(0.300)	(0.452)	
2013	$(0.013^{+1})$	(0.575)	(0.334)	(0.217)	
2014	(0.279) 0.594*	(0.373)	(0.301)	(0.409)	
2014	(0.324)	-0.003	(0.211)	(0.231)	
2015	(0.278)	(0.348)	(0.300)	(0.439) 0.212	
2015	(0.283)	-0.700	(0.210	(0.312)	
2016	(0.203)	(0.549)	0.313)	0.409	
2010	(0.281)	(0.561)	(0.316)	(0.464)	
2017	0.720**	-0.317	0.802**	0.559	
2011	(0.284)	(0.541)	(0.315)	(0.470)	
2018	0 779***	-0 444	0 787**	0.839*	
2010	(0.282)	(0.554)	(0.316)	(0.482)	
	(0.202)	(10.004)		(0.402)	
<b>T</b> '1' 1 1'1' ' A					
Initial conditions $t=0$					
Second job status					
Unly main job (reference)	0.005***	1 011***	9 1 69***	0.790***	
Marginal second job	$2.205^{-577}$	$1.011^{-4}$	$5.103^{\text{TT}}$	$2.(30^{+++})$	
Non movemal second :-1	(0.103 <i>)</i> 1 591***	(U.333) 1 667***	0.197)	(U.JUð) 5 156***	
non-marginal second Job	(0.224)	(0, 669)	(0.972)	0.100	
	(0.324)	(0.008)	(0.273)	(0.402)	

	Women		Men	
	Marginal	Non-marginal	Marginal	Non-marginal
	<u> </u>		 	
Initial conditions $t=0$				
Main job characteristics				
Main job monthly gross income				
up to 1st decile	-0.385	0.353	-0.758	-0.952
	(0.549)	(1.194)	(0.541)	(0.791)
1st to 2nd decile	-0.00606	0.631	0.358	-0.408
	(0.534)	(1.142)	(0.428)	(0.665)
2nd to 3rd decile	-0.252	0.530	-0.442	-0.836
	(0.526)	(1.125)	(0.409)	(0.604)
3rd to 4th decile	-0.513	0.411	-0.183	-1.434**
	(0.525)	(1.127)	(0.383)	(0.626)
4th to 5th decile	-0.281	0.557	-0.288	-0.939*
	(0.524)	(1.106)	(0.367)	(0.541)
5th to 6th decile	-0.828	0.589	-0.595*	-1.297**
	(0.517)	(1.095)	(0.352)	(0.543)
6th to 7th decile	-0.679	0.205	-0.531	-0.442
	(0.506)	(1.060)	(0.329)	(0.450)
7th to 8th decile	-0.593	0.477	-0.222	-0.387
	(0.505)	(1.044)	(0.306)	(0.403)
8th to 9th decile	-0.481	0.512	-0.117	0.145
	(0.497)	(1.053)	(0.283)	(0.340)
above 9th decile (reference)				
Tenure	0.000	0.040	0.0001	0.040
less than 2 years	-0.290	0.343	0.0331	0.342
	(0.213)	(0.538)	(0.253)	(0.410)
2-5 years	0.107	0.172	0.244	0.441
~ 40	(0.200)	(0.487)	(0.234)	(0.395)
5-10 years	-0.0865	0.180	-0.0829	-0.235
	(0.177)	(0.421)	(0.201)	(0.343)
more than 10 years (reference)				
Full-time $(1/0)$	0.234	0.582	0.170	0.625
$\mathbf{F}^{*}$ 1, $(1/0)$	(0.162)	(0.360)	(0.298)	(0.454)
Fixed-term contract $(1/0)$	-0.0856	$-0.559^{++}$	-0.185	(0.215)
Durbling survives and the $(1/0)$	(0.107)	(0.272)	(0.133)	(0.207)
Public service sector $(1/0)$	-0.0071	(0.422)	0.21(	(0.481)
Occuration	(0.179)	(0.422)	(0.274)	(0.373)
Untrained worker (reference)				
Untrained worker (reference)	0.0622	0 500	0.199	0.0100
Trailled worker	-0.0033	(0.399)	(0.227)	-0.0109
Foreman highly trained managerial duties	(0.193) 0.217	(0.432)	(0.337)	(0.371) 0.427
Foreman, inginy trained, managerial duties	(0.270)	(0.550)	(0.403)	(0.43)
Industry	(0.210)	(0.503)	(0.403)	(0.013)
$\Delta \operatorname{griculture}$ energy mining	0 321	1 007	0.473	-0.895
Agriculture, energy, mining	(0.605)	(1.445)	(0.527)	(0.757)
Manufacturing	0.673**	-0 530	0.0433	-0.551
Wanuacturing	(0.286)	(0.711)	(0.277)	(0.438)
Construction	1 873***	2 491**	-0.291	-0.766
Construction	(0.596)	(1.089)	(0.446)	(0.586)
Services	0.306	-0.662	-0.117	-0 589*
Firmsize	(0.219)	(0.439)	(0.255)	(0.353)
less than 20 employees (reference)		(0.100)		(0.000)
20 to 199 employees	-0.0300	0.302	0.0912	0.0366
20.00 100 employees	(0.177)	(0.409)	(0.246)	(0.400)
200 and more employees	-0.00904	0.145	0.0932	0.302
200 and more employees	(0.186)	(0.421)	(0.262)	(0.423)
		( /	· · · · · · · · · · · · · · · · · · ·	()

### 

	Women		Men	
	Marginal	Non-marginal	Marginal	Non-marginal
Initial conditions $t=0$				
Initial conditions l=0 Worker characteristics				
Family status				
Single and no children (reference)				
Partner and no children	0.0762	-0.465	-0.348	-0 108
i artifer and no enharch	(0.217)	(0.476)	(0.251)	(0.411)
Partner and children	(0.217) 0.326	(0.470) 0.144	(0.201)	(0.411) 0.244
I at their and children	(0.320)	(0.510)	(0.124)	(0.244)
Single and children	(0.252)	(0.513) 0.184	0.247)	(0.394) 0.007
Single and children	(0.300)	(0.104)	(0.527)	(1.025)
Desformed monting house first ich	(0.233)	(0.003)	(0.527)	(1.050)
Preferred working hours first job	0.0000	0.144	0.157	0.00179
More nours	-0.0008	-0.144	-0.10(	(0.00172)
	(0.117)	(0.209)	(0.140)	(0.223)
Same nours (reference)	0.005*	0.0200	0.0010	
Less hours	$0.205^{*}$	0.0398	0.0812	0.0442
	(0.124)	(0.281)	(0.151)	(0.255)
Employment status partner				
Not working (reference)				—
Full time	-0.0307	0.0136	0.288	-0.344
	(0.149)	(0.331)	(0.225)	(0.326)
Part time	0.456	-0.187	0.217	-0.377
	(0.391)	(0.963)	(0.197)	(0.314)
Other	-0.0153	0.886	0.830***	-0.775*
	(0.648)	(1.488)	(0.275)	(0.466)
Age				
18-25 years old	0.520	1.359	0.850	-0.138
	(0.515)	(1.148)	(0.621)	(1.038)
26-35 years old	0.414	1.091	0.627	0.103
	(0.425)	(0.950)	(0.487)	(0.750)
36-45 years old	0.229	0.883	0.236	-0.0910
	(0.360)	(0.752)	(0.404)	(0.611)
46-55 years old	0.0248	0.550	-0.225	-0.186
	(0.283)	(0.580)	(0.293)	(0.435)
older than 55 years (reference)	_			
Skill level				
Low skill	0.490	-0.971	-1.754*	-2.532*
	(0.742)	(1.294)	(0.924)	(1.521)
Medium skill (reference)	— ´		´	
High skill	0.168	-1.487	-1.182*	-0.457
0	(0.693)	(1.058)	(0.612)	(1.140)
Lives in East Germany $(1/0)$	0.170	0.00771	-0.156	-0.640
	(0.484)	(0.800)	(0.543)	(0.710)
Migration Background	()	(/)		()
No migration background (reference)		_		
Direct migration background	0.0124	0.00527	0.0286	0.0461
Direct ingration background	(0.471)	(0.604)	(0.553)	(0.513)
Indirect migration background	0.0302	0.00293	0.0108	0 00974
mancet ingration background	(0.285)	(0.00258)	(0.347)	(0.0046)
	(0.200)	(0.0200)	(0.347)	(0.0940)

	Women		Men	
	Marginal	Non-marginal	Marginal	Non-marginal
$Average  of  time{-}varying  variables$				
Main job characteristics				
Main job monthly gross income				
up to 1st decile	0.0896	-1.111	-1.239	-1.832
	(0.726)	(1.628)	(0.955)	(1.553)
1st to 2nd decile	-0.284	-0.996	-0.396	-2.256*
	(0.695)	(1.509)	(0.739)	(1.255)
2nd to 3rd decile	0.0417	-0.458	0.106	-1.030
	(0.684)	(1.443)	(0.673)	(1.094)
3rd to 4th decile	0.218	-0.470	0.111	-0.269
	(0.686)	(1.449)	(0.621)	(1.063)
4th to 5th decile	-0.130	0.00495	0.0188	-0.491
	(0.674)	(1.394)	(0.570)	(0.888)
5th to 6th decile	0.526	0.442	0.548	-0.336
	(0.670)	(1.359)	(0.557)	(0.834)
6th to 7th decile	1.142*	-0.0461	0.356	-0.0248
	(0.629)	(1.358)	(0.515)	(0.736)
7th to 8th decile	-0.0579	0.558	0.263	-0.915
	(0.629)	(1.199)	(0.478)	(0.656)
8th to 9th decile	0.632	0.238	-0.382	-1.074*
	(0.612)	(1.307)	(0.430)	(0.593)
above 9th decile				
Tenure	o 10 <b>-</b>	0.40 <b>×</b>	0.001	0.001
less than 2 years	0.437	0.485	0.221	0.0917
2.5	(0.336)	(0.842)	(0.411)	(0.687)
2-5 years	0.380	1.147	-0.398	0.683
5 10	(0.308)	(0.703)	(0.391)	(0.680)
5-10 years	0.454	0.127	0.289	-0.108
	(0.291)	(0.676)	(0.358)	(0.053)
more than 10 years $E_{\rm rel}$ time (1/0)	0.155	0.110	0 5 4 9	1.014
Full-time $(1/0)$	-0.100	(0.110)	(0.472)	-1.014
Exact target contract $(1/0)$	(0.209)	(0.371)	(0.472)	(0.724) 0.217
Fixed-term contract $(1/0)$	(0.268)	(0.622)	-0.00100	(0.317)
Public corrigo soster $(1/0)$	(0.208)	(0.078)		(0.400)
1  ublic Service Sector  (1/0)	(0.281)	(0.714)	(0.373)	-0.550)
Occupation	(0.201)	(0.714)	(0.515)	(0.000)
Untrained worker	3 488	2557	-2.067	9 094
Chitranica worker	(5,385)	(6.934)	(4.033)	(7,771)
Trained worker	(3.300) 3 402	(0.354)	-1 847	10.32
Fighted worker	(5,370)	(6.903)	$(4\ 000)$	(7749)
Foreman highly trained managerial duties	3 732	(0.000)	-2 074	10 72
roromani, inging trainoù, managoriar dutios	(5.380)	(6.949)	(4.006)	(7.727)
Industry	(0.000)	(010 10)	(1.000)	()
Agriculture, energy, mining	-1.179	-1.178	-1.187	2.447
0 ,	(1.144)	(2.636)	(1.013)	(1.623)
Manufacturing	-0.955	-0.842	-0.0508	1.383
	(0.655)	(1.637)	(0.775)	(1.443)
Construction	-1.640*	-21.96***	0.338	1.923
	(0.975)	(3.248)	(0.888)	(1.574)
Services	-0.244	0.345	0.631	$2.638^{*}$
	(0.607)	(1.357)	(0.757)	(1.433)

	Women		Men	
	Marginal	Non-marginal	Marginal	Non-marginal
Average of time-varying variables Main job characteristics Firmsize				
less than 20 employees	0.239	-0.120	0.618	0.750
1 0	(0.297)	(0.724)	(0.423)	(0.632)
20 to $199$ employees	0.0149	-0.413	0.0537	0.375
1.0	(0.247)	(0.538)	(0.287)	(0.468)
200 and more employees Worker characteristics				
Family status	0.969	0.459	0 575	0.007**
Single and no children	(0.302)	-0.455	-0.373	
	(0.379)	(0.834)	(0.913)	(1.517)
Partner and no children	0.510	0.792	-0.299	-2.836*
	(0.410)	(0.849)	(0.954)	(1.550)
Partner and children	0.169	0.964	-0.764	-2.678*
	(0.389)	(0.819)	(0.956)	(1.574)
Single and children				
Preferred working hours first job	0.000	0.100	0.01.0*	
More hours	-0.223	0.130	-0.916*	-1.285*
	(0.452)	(0.983)	(0.495)	(0.712)
Same hours	-1.120**	-1.640*	-1.434***	-1.311*
	(0.456)	(0.978)	(0.492)	(0.702)
Less hours	-0.967**	-0.762	$-1.510^{***}$	-1.893***
	(0.438)	(0.942)	(0.481)	(0.685)
Employment status partner				
Not working	-0.637	2.857	0.251	-1.196
	(1.049)	(2.816)	(0.447)	(0.728)
Full time	-0.742	2.317	0.430	-0.932
	(1.049)	(2.838)	(0.495)	(0.770)
Part time	-0.549	1.849	0.133	-1.015
	(1.133)	(3.187)	(0.469)	(0.725)
Other		_	—	
Age				
18-25 years old	-0.808	0.0453	-0.798	-1.082
	(0.767)	(2.087)	(0.915)	(2.051)
26-35 years old	-0.0866	0.0280	-0.349	0.0407
	(0.547)	(1.319)	(0.629)	(1.028)
36-45 years old	0.136	0.0271	-0.0443	-0.626
v	(0.429)	(0.946)	(0.496)	(0.790)
46-55 years old	0.288	0.212	-0.0424	-0.837
v	(0.327)	(0.704)	(0.370)	(0.609)
older than 55 years				
Skill level				
Low skill	-1.638	1.737	1.500	$6.284^{***}$
	(1.220)	(2.313)	(1.370)	(1.578)
Medium skill	1.414	0.487	-1.178	1.052
	(1.142)	(1.576)	(0.952)	(1.556)
High skill				(1.000)
Lives in East Germany $(1/0)$	-0.658	3.220***	1.976**	1,230
$\mathbf{L}_{\mathbf{r},\mathbf{r},\mathbf{r},\mathbf{r},\mathbf{r},\mathbf{r},\mathbf{r},\mathbf{r}$	(0.929)	$(1\ 129)$	(0.942)	(1.200)
	(0.323)	(1.123)	(0.342)	(1.202)

### References

- Allen, W. D. (1998), 'The moonlighting decision of unmarried men and women: Family and labor market influences', Atlantic Economic Journal, 26(2): 190–205.
- Amuedo-Dorantes, C. and Kimmel, J. (2009), 'Moonlighting over the Business Cycle', *Economic Inquiry*, 47(4): 754–765.
- Auray, S., Fuller, D. L., and Vandenbroucke, G. (2021), 'Comparative advantage and moonlighting', *European Economic Review*, 139: 103897.
- Averett, S. L. (2001), 'Moonlighting: multiple motives and gender differences', Applied Economics, 33(11): 1391–1410.
- Boheim, R. and Taylor, M. P. (2004), 'And in the Evening She's a Singer with the Band Second Jobs, Plight or Pleasure?', *SSRN Electronic Journal*.
- Choe, C., Oaxaca, R. L., and Renna, F. (2017), 'Constrained vs unconstrained labor supply: the economics of dual job holding', *Journal of Population Economics*, 31(4): 1279–1319.
- Conway, K. S. and Kimmel, J. (1998), 'Male labor supply estimates and the decision to moonlight', *Labour Economics*, 5(2): 135–166.
- Dickey, H., Watson, V., and Zangelidis, A. (2011), 'Is it all about money? An examination of the motives behind moonlighting', *Applied Economics*, 43(26): 3767–3774.
- Goebel, J., Grabka, M. M., Liebig, S., Kroh, M., Richter, D., Schröder, C., and Schupp, J. (2018), 'The German Socio-Economic Panel (SOEP)', Jahrbücher für Nationalökonomie und Statistik, 239(2): 345–360.
- Guthrie, H. (1969), 'Teachers in the moonlight', Monthly Labour Review, 92: 28–31.
- Hamel, H. (1967), 'Moonlighting An economic phenomenon', *Monthly Labour Review*, 90: 17–22.
- Heineck, G. (2009), 'The determinants of secondary jobholding in Germany and the UK', Zeitschrift fuer ArbeitsmarktForschung, 42(2): 107–120.
- Hirsch, B. T., Husain, M. M., and Winters, J. V. (2017), 'The Puzzling Pattern of Multiple Job Holding across U.S. Labor Markets', Southern Economic Journal, 84(1): 26–51.
- Kimmel, J. and Conway, K. S. (2001), 'Who Moonlights and Why? Evidence from the SIPP', Industrial Relations: A Journal of Economy and Society, 40(1): 89–120.
- Klinger, S. and Weber, E. (2020), 'Secondary job holding in Germany', *Applied Economics*, 52(30): 3238–3256.
- Panos, G. A., Pouliakas, K., and Zangelidis, A. (2014), 'Multiple Job Holding, Skill Diversification, and Mobility', *Industrial Relations: A Journal of Economy and Society*, 53(2): 223–272.
- Perlman, R. (1966), 'Observations on Overtime and Moonlighting', Southern Economic Journal, 33(2): 237.

- Pouliakas, K. (2017), 'Multiple job-holding: Career pathway or dire straits?', *IZA World of Labor*.
- Rabe-Hesketh, S. and Skrondal, A. (2013), 'Avoiding biased versions of Wooldridge's simple solution to the initial conditions problem', *Economics Letters*, 120(2): 346–349.
- Renna, F. (2006), 'Moonlighting and overtime: A cross-country analysis', Journal of Labor Research, 27(4): 575–591.
- Schmidt, T. and Voss, D. (2014), 'Arbeitsmarkt- und geschlechtsdifferenzielle Einflussfaktoren für die Ausübung einer geringfügigen Nebenbeschäftigung', Industrielle Beziehungen. Zeitschrift für Arbeit, Organisation und Management, (1): 36–57.
- Shishko, R. and Rostker, B. (1976), 'The Economics of Multiple Job Holding.', *The American Economic Review*, 66(3): 298–308.
- Wooldridge, J. M. (2005), 'Simple solutions to the initial conditions problem in dynamic, nonlinear panel data models with unobserved heterogeneity', *Journal of Applied Econometrics*, 20(1): 39–54.
- Wu, Z., Baimbridge, M., and Zhu, Y. (2009), 'Multiple job holding in the United Kingdom: evidence from the British Household Panel Survey', *Applied Economics*, 41(21): 2751–2766.

# **Working Paper Series in Economics**

(recent issues)

- No. 415 *Joachim Wagner*: Online Channels Sales Premia in Times of COVID-19: First Evidence from Germany, November 2022
- No. 414 Boris Hirsch, Elke J. Jahn, Alan Manning, and Michael Oberfichtner: The wage elasticity of recruitment, Oktober 2022
- No. 413 *Lukas Tohoff and Mario Mechtel*: Fading Shooting Stars The Relative Age Effect, Misallocation of Talent, and Returns to Training in German Elite Youth Soccer, September 2022
- No. 412 Joachim Wagner: The first 50 contributions to the Data Observer Series An overview, May 2022
- No. 411 *Mats Petter Kahl and Thomas Wein*: How to Reach the Land of Cockaigne? Edgeworth Cycle Theory and Why a Gasoline Station is the First to Raise Its Price, April 2022
- No. 410 *Joachim Wagner:* Website premia for extensive margins of international firm activities Evidence for SMEs from 34 countries; April 2022
- No. 409 *Joachim Wagner:* Firm survival and gender of firm owner in times of COVID-19 Evidence from 10 European countries, March 2022
- No. 408 *Boris Hirsch, Philipp Lentge and Claus Schnabel*: Uncovered workers in plants covered by collective bargaining: Who are they and how do they fare?, February 2022
- No. 407 *Lena Dräger, Michael J. Lamla and Damjan Pfajfar:* How to limit the Spillover from the 2021 Inflation Surge to Inflation Expectations?, February 2022
- No. 406 Institut für Volkswirtschaftslehre: Forschungsbericht 2021, Januar 2022
- No. 405 *Leif Jacobs, Lara Quack and Mario Mechtel:* Distributional Effects of Carbon Pricing by Transport Fuel Taxation, December 2021
- No. 404 *Boris Hirsch and Philipp Lentge:* Non-Base Compensation and the Gender Pay Gap, July 2021
- No. 403 *Michael J. Lamla and Dmitri V. Vinogradov:* Is the Word of a Gentleman as Good as His Tweet? Policy communications of the Bank of England, May 2021
- No. 402 *Lena Dräger, Michael J. Lamla and Damjan Pfajfar:* The Hidden Heterogeneity of Inflation and Interest Rate Expectations: The Role of Preferences, May 2021
- No. 401 *Joachim Wagner:* The Good have a Website Evidence on website premia for firms from 18 European countries, April 2021
- No. 400 *Luise Görges:* Of housewives and feminists: Gender norms and intra-household division of labour, April 2021
- No. 399 *Joachim Wagner:* With a little help from my website. Firm survival and web presence in times of COVID-19 Evidence from 10 European countries, April 2021
- No. 398 *Katja Seidel:* The transition from School to Post-Secondary Education What factors affect educational decisions?, March 2021

- No. 397 Institut für Volkswirtschaftslehre: Forschungsbericht 2020, Januar 2021
- No. 396 Sabien Dobbelaere, Boris Hirsch, Steffen Mueller and Georg Neuschaeffer: Organised Labour, Labour Market Imperfections, and Employer Wage Premia, December 2020
- No. 395 *Stjepan Srhoj, Vanja Vitezić and Joachim Wagner:* Export boosting policies and firm behaviour: Review of empirical evidence around the world, November 2020
- No. 394 *Thomas Wein:* Why abandoning the paradise? Stations incentives to reduce gasoline prices at first, August 2020
- No. 393 Sarah Geschonke and Thomas Wein: Privacy Paradox Economic Uncertainty Theory and Legal Consequences, August 2020
- No. 392 *Mats P. Kahl:* Impact of Cross-Border Competition on the German Retail Gasoline Market – German-Polish Border, July 2020
- No. 391 *John P. Weche and Joachim Wagner:* Markups and Concentration in the Context of Digitization: Evidence from German Manufacturing Industries, July 2020
- No. 390 Thomas Wein: Cartel behavior and efficient sanctioning by criminal sentences, July 2020
- No. 389 *Christoph Kleineberg:* Market definition of the German retail gasoline industry on highways and those in the immediate vicinity, July 2020
- No. 388 Institut für Volkswirtschaftslehre: Forschungsbericht 2019, Januar 2020
- No. 387 *Boris Hirsch, Elke J. Jahn, and Thomas Zwick:* Birds, Birds, Birds: Co-worker Similarity, Workplace Diversity, and Voluntary Turnover, May 2019
- No. 386 *Joachim Wagner:* Transaction data for Germany's exports and imports of goods, May 2019
- No. 385 Joachim Wagner: Export Scope and Characteristics of Destination Countries: Evidence from German Transaction Data, May 2019
- No. 384 *Antonia Arsova:* Exchange rate pass-through to import prices in Europe: A panel cointegration approach, February 2019
- No. 383 Institut für Volkswirtschaftslehre: Forschungsbericht 2018, Januar 2019
- No. 382 *Jörg Schwiebert*: A Sample Selection Model for Fractional Response Variables, April 2018
- No. 381 Jörg Schwiebert: A Bivarate Fractional Probit Model, April 2018
- No. 380 *Boris Hirsch and Steffen Mueller:* Firm wage premia, industrial relations, and rent sharing in Germany, February 2018
- No. 379 *John P. Weche and Achim Wambach:* The fall and rise of market power in Europe, January 2018
- No.378: Institut für Volkswirtschaftslehre: Forschungsbericht 2017, Januar 2018
- No.377: Inna Petrunyk and Christian Pfeifer: Shortening the potential duration of unemployment benefits and labor market outcomes: Evidence from a natural experiment in Germany, January 2018

- No.376: *Katharina Rogge, Markus Groth und Roland Schuhr:* Offenlegung von CO2-Emissionen und Klimastrategien der CDAX-Unternehmen eine statistische Analyse erklärender Faktoren am Beispiel der CDP-Klimaberichterstattung, Oktober 2017
- No.375: *Christoph Kleineberg und Thomas Wein:* Verdrängungspreise an Tankstellen?, September 2017
- No.374: *Markus Groth, Laura Schäfer und Pia Scholz*: 200 Jahre "On the Principles of Political Economy and Taxation" Eine historische Einordnung und Würdigung, März 2017
- No.373: *Joachim Wagner:* It pays to be active on many foreign markets Profitability in German multi-market exporters and importers from manufacturing industries, March 2017
- No.372: *Joachim Wagner:* Productivity premia for many modes of internationalization A replication study of Békes / Muraközy, Economics Letters (2016), March 2017 [published in: International Journal for Re-Views in Empirical Economics IREE, Vol. 1 (2017-4)]
- No.371: *Marius Stankoweit, Markus Groth and Daniela Jacob:* On the Heterogeneity of the Economic Value of Electricity Distribution Networks: an Application to Germany, March 2017
- No.370: *Joachim Wagner:* Firm size and the use of export intermediaries. A replication study of Abel-Koch, The World Economy (2013), January 2017 [published in: International Journal for Re-Views in Empirical Economics IREE, Vol. 1 (2017-1)]
- No.369: *Joachim Wagner:* Multiple import sourcing First evidence for German enterprises from manufacturing industries, January 2017 [published in : Open Economies Review 29 (2018), 1, 165-175]
- No.368: *Joachim Wagner:* Active on many foreign markets A portrait of German multi-market exporters and importers from manufacturing industries, January 2017 [published in: Jahrbücher für Nationalökonomie und Statistik 238 (2018), 2, 157-182]
- No.367: Institut für Volkswirtschaftslehre: Forschungsbericht 2016, Januar 2017
- No.366: *Tim W. Dornis and Thomas Wein:* Trademarks, Comparative Advertising, and Product Imitations: An Untold Story of Law and Economics, September 2016
- No.365: *Joachim Wagner:* Intra-good trade in Germany: A first look at the evidence, August 2016 [published in: Applied Economics 49 (2017), 57, 5753-5761]
- No.364: *Markus Groth and Annette Brunsmeier:* A cross-sectoral analysis of climate change risk drivers based on companies' responses to the CDP's climate change information request, June 2016
- No.363: Arne Neukirch and Thomas Wein: Collusive Upward Gasoline Price Movements in Medium-Sized German Cities, June 2016
- No.362: *Katja Seidel:* Job Characteristics and their Effect on the Intention to Quit Apprenticeship., May 2016
- No.361: *Katja Seidel:* Apprenticeship: The Intention to Quit and the Role of Secondary Jobs in It., May 2016
- No.360: *Joachim Wagner:* Trade costs shocks and lumpiness of imports: Evidence from the Fukushima disaster, May 2016 [published in: Economics Bulletin 37 (2017), 1, 149-155]

(see www.leuphana.de/institute/ivwl/working-papers.html for a complete list)

Leuphana Universität Lüneburg Institut für Volkswirtschaftslehre Postfach 2440 D-21314 Lüneburg Tel.: ++49 4131 677 2321 email: christina.korf@leuphana.de

www.leuphana.de/institute/ivwl/working-papers.html